

Clinical Commissioning Policy: Temperaturecontrolled laminar airflow device for persistent allergic asthma (children)

Reference: NHS England: 16013/P



NHS England INFORMATION READER BOX

Directorate		
Medical	Operations and Information	Specialised Commissioning
Nursing	Trans. & Corp. Ops.	Commissioning Strategy
Finance		

Publications Gateway Reference: 05527s			
Document Purpose	Policy		
Document Name	Clinical Commissioning Policy 16013/P		
Author	Specialised Commissioning Team		
Publication Date	13 July 2016		
Target Audience	CCG Clinical Leaders, Care Trust CEs, Foundation Trust CEs, Medical Directors, Directors of PH, Directors of Nursing, NHS England Regional Directors, NHS England Directors of Commissioning Operations, Directors of Finance, NHS Trust CEs		
Additional Circulation List			
Description	Not Routinely Commissioned - NHS England will not routinely commission this specialised treatment in accordance with the criteria described in this policy.		
Cross Reference	N/A		
Superseded Docs (if applicable)	N/A		
Action Required	N/A		
Timing / Deadlines (if applicable)	N/A		
Contact Details for further information	england.specialisedcommissioning@nhs.net		

Document Status

This is a controlled document. Whilst this document may be printed, the electronic version posted on the intranet is the controlled copy. Any printed copies of this document are not controlled. As a controlled document, this document should not be saved onto local or network drives but should always be accessed from the intranet.

Clinical Commissioning Policy: Temperature-controlled laminar airflow device for persistent allergic asthma (children)

First published: July 2016

Prepared by NHS England Specialised Services Clinical Reference Group for Paediatric Medicine

Published by NHS England, in electronic format only.

Contents

1	Introduction	7
2	Definitions	7
3	Aims and Objectives	7
4	Epidemiology and Needs Assessment	8
5	Evidence Base	8
6	Documents which have informed this Policy	11
7	Date of Review	11
Refe	erences	12

Policy Statement

NHS England will not routinely commission the temperature-controlled laminar airflow device for persistent allergic asthma (children) in accordance with the criteria outlined in this document. In creating this policy NHS England has reviewed this clinical condition and the options for its treatment. It has considered the place of this treatment in current clinical practice, whether scientific research has shown the treatment to be of benefit to patients, (including how any benefit is balanced against possible risks) and whether its use represents the best use of NHS resources. This policy document outlines the arrangements for funding of this treatment for the population in England.

Equality Statement

Promoting equality and addressing health inequalities are at the heart of NHS England's values. Throughout the development of the policies and processes cited in this document, we have:

- Given due regard to the need to eliminate discrimination, harassment and victimisation, to advance equality of opportunity, and to foster good relations between people who share a relevant protected characteristic (as cited under the Equality Act 2010) and those who do not share it; and
- Given regard to the need to reduce inequalities between patients in access to, and outcomes from healthcare services and to ensure services are provided in an integrated way where this might reduce health inequalities

Plain Language Summary

About persistent allergic asthma

Asthma is characterised by signs or symptoms including breathlessness, chest tightness, wheezing, and coughing. In allergic asthma, symptoms are caused by an allergic reaction. Common items that cause allergic asthma include:

- pollen
- pets
- house dust mites.

About the new treatment

Long-term allergic asthma may be poorly controlled, despite the patient having high intensity treatment. Such patients are likely to be looked after by a hospital doctor who specialises in breathing problems.

New treatments called 'temperature-controlled laminar airflow devices' can be used for these patients to use at home.

- The device works at the patient's bedside while the patient sleeps.
- It draws air from the room through a filter.
- The air is then cooled to below the temperature in the bedroom and then slowly released into the air in the bedroom.
- As the cooled air is heavier than the surrounding air, it settles down to where the patient is breathing.

This means the device provides filtered air around the patient's face throughout the night. It does this without creating a draught or dehydration.

What we have decided

NHS England has carefully reviewed the evidence to treat allergic asthma with temperature-controlled laminar airflow devices. We have concluded that there is not enough evidence to make the treatment available at this time.

1 Introduction

This document describes the evidence that has been considered by NHS England in formulating a proposal to not routinely commission temperature-controlled laminar airflow devices for the treatment of persistent allergic asthma in children.

The UK direct healthcare costs for asthma are over £1 billion per year, of which a high percentage is focused on those with the top quartile of severity-related drug requirements. Such patients require specialist tertiary investigation and care. Detailed assessment is required to establish which patients have truly therapy resistant disease compared with those that have potentially avoidable contributors to high morbidity, such as: poor concordance with therapy, significant co-morbidities (such as obesity, rhinitis, etc.), avoidable adjuvants such as cigarette smoke, wrong prescription or wrong diagnosis. Having excluded the former, those with severe persistent allergic asthma are considered for treatment with omalizumab.

This policy has considered the clinical evidence available to support the routine commissioning of temperature-controlled laminar airflow devices for children suffering from persistent allergic asthma as a treatment prior to the consideration of omalizumab.

2 Definitions

Temperature-controlled laminar airflow devices are a method of stringent aeroallergen avoidance by controlling the air quality around a patient's breathing zone throughout the night.

Persistent allergic asthma in children is defined as those who meet the.

3 Aims and Objectives

This policy aims to define the current commissioning proposition for Temperature-Global Initiative for Asthma (GINA) 2002 step 4 criteria, requiring referral for specialist advice controlled Laminar Airflow based on the current evidence base.

The objective is to ensure evidence based commissioning in the use of temperaturecontrolled laminar airflow device for the treatment of persistent allergic asthma in children.

4 Epidemiology and Needs Assessment

The quality and outcomes framework (2008) estimated that 5.9% of the UK population have asthma, with estimates ranging from 3 to 5.4 million.

Asthma UK estimated that between 2008 and 2009 there were 79,794 emergency hospital admissions for asthma in England, of which 30,740 were of children aged up to 14 years.

According to Asthma UK, 75% of all hospital admissions for asthma are avoidable through good asthma management and routine care.

5 Evidence Base

This document describes the evidence that has been considered by NHS England in formulating a proposal to not routinely commission temperature-controlled laminar airflow devices for the treatment of persistent allergic asthma in children.

The evidence review looked to consider the following research questions:

Are temperature-controlled laminar airflow (TCLA) devices clinically effective in reducing airway inflammation, sustaining improved asthma control, reducing annual exacerbation rates, and improving quality of life patients with persistent allergic asthma compared with no intervention or with other standardised treatments?

Are TCLA devices cost effective in children with persistent allergic asthma?

Based on the inclusion and exclusion criteria detailed in the appendix, 4 studies were selected for full review. This includes a National Institute for Health and Care Excellence (NICE) Medtech Innovation Briefing published in August 2014. There are two Grade 1- studies, including a randomised controlled trial and one cost-

effectiveness study which were both funded by Airsonnet AB. There is one Grade 3 case series which addresses clinical effectiveness in addition to quality of life outcomes.

The best current evidence for use of TCLA devices for persistent allergic asthma comes from a single, relatively large, randomised study (Boyle et al 2012). This study was not designed primarily to evaluate the clinical effectiveness of TCLA such as the impact on asthma exacerbations, hospitalisation, emergency room visits and use of medication. The history of frequent or severe exacerbations was not an inclusion criterion. In the included study population, the active and placebo groups showed no statistically significant difference in standard asthma medication use and asthma exacerbations. There was no follow-up of the patients post study period to evaluate long term effectiveness. TCLA treatment was associated with a greater decrease in fraction exhaled nitric oxide (FeNO) than placebo during the study period of one year. There was no significant impact on blood eosinophil counts, total IgE level and overall lung function between treatment groups.

Despite the identified limitations in study design, there is some evidence from the study that TCLA can improve quality of life for patients with more severe and uncontrolled asthma.

Of those patients who had at least 1 day of treatment with the Airsonett device, there was a significantly greater proportion with increase in AQLQ score of at least 0.5 points and 1 point compared with the placebo group. Statistically significant improvements using this measure were most noticeably reported in those with poor symptom control (ACT<18) who received high intensity treatment. These differences in improvement of quality of life only reach statistical significance in the subgroup of patients aged below 12 years. The study was powered on subgroup> 12 years.

The other study which specifically addressed questions of clinical effectiveness in addition to quality of life (Schauer et al 2015) is a non-randomized uncontrolled pre and post retrospective observational study to investigate the effect of 12 months' TCLA use in a population of 30 patients (27 finished full 12 month follow-up). Due to

the small number of patients and the observational study design, the findings from the study cannot be generalised to a broader patient population.

Medtech innovation briefing on the Airsonett temperature-controlled laminar airflow device for persistent allergic asthma (NICE, 2014) advises that the device is non-invasive and non-pharmaceutical. No treatment related adverse events have been identified. Two trials (Boyle et al 2012, Pedroletti et al. 2009) showed statistically significant improvement in asthma related quality of life in people with severe persistent allergic asthma when Airsonett was compared with a placebo device. There was no statistically significant difference in asthma medication usage or exacerbation rates, which were secondary outcome measures in one randomised controlled trial. The second Randomised Control Trial (Pedroletti et al. 2009) was identified as a crossover study with a very small sample size, and no details were reported on the methods of randomisation or blinding. All other studies reviewed had small sample sizes and provided insufficient information to assess their quality.

There is currently limited published evidence on how the use of the Airsonett device, or similar TCLA would affect NHS resources by either reducing the use of omalizumab and other alternative treatment options or reducing asthma exacerbations.

The Medtech review advises that the average cost of long term treatment with Airsonett is £5.72 per patient per day. The estimated cost of an add on therapy currently used in NHS practice, omalizumab, is £23 per day.

The only study on cost-effectiveness of TCLA (Brodtkorb et al 2010) is based on Markov model of QALYs for next 5 year using data from Pedroletti et al (2011). The study concludes that Airshower strategy could result in a mean gain of 0.25 QALYs per patient in Sweden, thus yielding an approximate cost per QALY gained of under £25,571 as long as the cost of Airshower is below £5991 [Original figures provided in euros and converted to the nearest full pound based on conversion rate on 19/10/2015 of £1 to 1.37 euros and is provided as a guideline for comparison only]. The study does not include comparative cost effectiveness with existing comparator

interventions such as omalizumab, immunosuppressant therapy and bronchial thermoplasty.

The UK LASER Trial (Laminar Airflow in Severe Asthma for Exacerbation Reduction) currently underway could provide evidence regarding the clinical and comparative cost effectiveness of TCLA in patients with persistent allergic asthma.

6 Documents which have informed this Policy

National Institute for Health and Care Excellence Medtech Innovation Briefing Report. The Airsonnet Temperature-Controlled Laminar Airflow Device for persistent allergic asthma. 2014.

National Institute for Health and Care Excellence Technology Appraisal Guidance 278 Omalizumab for treating severe persistent allergic asthma. 2013.

7 Date of Review

This document will be reviewed when information is received which indicates that the policy requires revision.

References

Asthma UK for Journalists: Key facts and stats: www.asthma.uk.org

Boyle Robert, J; Pedroletti, Christophe; Wickman, Magnus; Bjermer, Leif; Valovirta, Erikka; Dahl, Ronald; Von Bern, Andrea; Zetterstrom, Olof; Warner, John. 4A Study Group. Nocturnal temperature controlled laminar airflow for treating atopic asthma: a randomised controlled trial. Thorax 2012.

Brodtkorb, Thor-Henrik; Zetterstrom, Olle; Tinghog, Gustav. Cost effectiveness of clean air administered to the breathing zone in allergic asthma. Clin Respir J. 2010.

Schauer, Uwe; Bergmann Karl-Christian; Gerrstaluer, Michael; Lehmann, Sylvia; Gappa, Monika; Brenneken, Amelia; Schulz, Christian; Ahrens, Peter; Schieber, Jens; Wittmann, Michael; Hamelmann, Eckard and all the members of the German Asthma Net (GAN). Improved asthma control in patients with severe persistent allergic asthma after 12 months of nightly temperature-controlled laminar airflow: an observational study with retrospective comparisons. 0. 2015.

Pedroletti C, Millinger E, Dahlen B et al (2009) Clinical effects of purified air administered to breathing zone in allergic asthma: a double –blind randomised cross over trial. Respiratory Medicine 103 (9), 1313-19